

# Temi di Discussione

(Working Papers)

Venture capitalists at work: what are the effects on the firms they finance?

by Raffaello Bronzini, Gianpaolo Caramellino and Silvia Magri







# Temi di discussione

(Working papers)

Venture capitalists at work: what are the effects on the firms they finance?

by Raffaello Bronzini, Gianpaolo Caramellino and Silvia Magri

Number 1131 - September 2017

The purpose of the Temi di discussione series is to promote the circulation of working papers prepared within the Bank of Italy or presented in Bank seminars by outside economists with the aim of stimulating comments and suggestions.

The views expressed in the articles are those of the authors and do not involve the responsibility of the Bank.

*Editorial Board:* Ines Buono, Marco Casiraghi, Valentina Aprigliano, Nicola Branzoli, Francesco Caprioli, Emanuele Ciani, Vincenzo Cuciniello, Davide Delle Monache, Giuseppe Ilardi, Andrea Linarello, Juho Taneli Makinen, Valerio Nispi Landi, Lucia Paola Maria Rizzica, Massimiliano Stacchini. *Editorial Assistants:* Roberto Marano, Nicoletta Olivanti.

ISSN 1594-7939 (print) ISSN 2281-3950 (online)

Printed by the Printing and Publishing Division of the Bank of Italy

## VENTURE CAPITALISTS AT WORK: WHAT ARE THE EFFECTS ON THE FIRMS THEY FINANCE?

by Raffaello Bronzini°, Gianpaolo Caramellino^ and Silvia Magri\*

#### Abstract

Italian startups financed by venture capitalists (VCs) experience a faster growth in size and become more innovative compared with other startups. VC-backed firms also show a much larger increase in equity and a reduction in their leverage. This evidence is obtained by comparing a representative sample of firms financed by private VCs in the period 2004-2014 with a sample of firms rejected by VC at the very last stage of the screening process or in the due diligence phase. These firms narrowly lost the contest and before VC financing have very similar observable and unobservable characteristics to the VC-backed firms; self-selection is specifically taken into account. The effects on firms' size and innovation are not exclusively explained by equity financing. The results hold when we restrict the comparison to firms in the control group that also increase their equity from investors other than VCs: this suggests that VC effects can also be linked to their managerial expertise and network connections. Finally, the results are exclusively driven by independent VC investors compared with captive VCs.

## JEL Classification: G21, G24, G32, O30.

Keywords: venture capital, innovation, firm financial structure, differences-in-differences.

#### Contents

1. Introduction	5
2. Literature review	9
3. Selection process among venture capitalists and research design	11
4. Source of data and descriptive statistics	13
5. Empirical strategy	15
6. Results of the effects of VC financing	17
7. Robustness and extensions of the results	20
7.1 Comparison with the firms in the control sample that increased their equity	20
7.2 Captive and independent venture capitalists	21
8. Discussion of results and conclusions	22
References	25
Tables and figures	27

<sup>°</sup> Bank of Italy, Economic Research Division, Rome Branch.

<sup>^</sup> London School of Economics.

<sup>\*</sup> Bank of Italy, Directorate General for Economics, Statistics and Research.

# **1** Introduction <sup>1</sup>

Venture capital (VC) investors provide equity capital to early-stage, high growth potential startup companies that develop a new technology or a new business model in high-tech industries. Equity is an important source of finance for startup innovative companies that could find it difficult to obtain debt as banks normally require collateral they might lack of; additionally, debt financing involves the ability to service debt, while startup firms might not generate any cash flow for the initial years of activity. Venture capitalists aim at getting a return by selling their shares in the companies through a trade-sale or an Initial Public Offering (IPO). They usually expect important returns on some of their investments to offset the fact that a good amount of their projects will fail.<sup>2</sup> In order to increase the return of the investments, VC investors adopt an active form of financing: almost all of them sit on the board of directors and they provide entrepreneurs with advice and contacts.

VC investors might therefore have important effects on the firms they finance, whose performances are hence expected to be better than those of other similar firms that did not receive VC finance (Gompers and Lerner, 2001). This is not just because the equity capital they provide helps reducing the funding gap of high-tech startup firms, but also due to the fact that VC managerial and financial experience could be very useful in enhancing firms' grow perspectives. Finally they can also improve firms' performances through their network connections and a signaling effect on other financiers, specifically banks. On the other hand, following the VC intervention, important conflicts can arise in the governance of the firms, which could be harmful for their performances. First, the aims and strategies of VC investors could be very different from those of the entrepreneurs; specifically, most VC investors could have too a short-term investment perspective compared with that of the entrepreneurs, who can consider this feature detrimental for long-term firm performances. Although VC investors are committed to a company for a long haul, their primary aim is to find a

<sup>&</sup>lt;sup>1</sup>The views expressed in this article are those of the authors and do not necessarily reflect those of the Bank of Italy. Cristiana Rampazzi and Stefania De Mitri provided excellant research assistance. We would like to thank for their useful comments participants at the BOI and LSE-PhD seminars, at the 14 International Conference on Credit Risk Evaluation (Venice, October 2015) and at the 65 Midwest Finance Association (Atlanta, March 2016), specifically Matteo Benetton, Andreas Ek, Giorgio Gobbi, Juanita Gonzalez-Uribe, Andrea Lamorgese, Andrea Linarello, Francesca Lotti, Samuele Murtino, Daniel Paravisini, Enrico Rettore, Paolo Sestito, Enrico Sette, Luigi Federico Signorini, Roger Stein and Konstantinos Tokis. We are also grateful to the Italian Association of Private Equity and Venture Capital (AIFI) for their help in collecting information on rejected businesses by venture capitalists, and to Diana Del Colle who initially helped us with the dataset.

 $<sup>^{2}</sup>$ Shikhar Ghosh of Harvard Business School (HBS) found that three-quarters of US startups backed by venture capital failed to return the capital invested in them, let alone generate a positive return; the figure was calculated on a sample of 2,000 companies that received VC funding between 2004 and 2010. Entrepreneurs anonymous, The Economist, Sept 20th 2014.

good form of exit from the company.<sup>3</sup> Secondly, appropriability problems can arise as VC investors might just try to capture the innovative idea of the entrepreneurs and exploit it by themselves. The evaluation of VC effects is therefore an empirical question. As a matter of fact, some studies have found no or weak beneficial effects of VC investors on firms' results.

The aim of this paper is to evaluate the private VC contributions on Italian startups they finance. We focus on Italy where the VC market is still very underdeveloped compared with other European countries and the US (Figure 1). The evaluation of private VC activity is very important in our country where a public support has been suggested in order to provide a kick off to the expansion of the private VC market: with this purpose, some VC funds have been created, partially funded with public money.<sup>4</sup>

The most important challenge in this type of analysis is finding an adequate identification strategy so that a VC treatment effect is detectable, while selection effect is controlled for. Firms that apply for VC funding may be different: the decision to apply may be related to the quality of the new idea and the consequent determination to exploit it. Moreover, VC investors could be smart enough to select the best startup high-tech companies. In other words, there could exist some firm unobservable features (unobservable to the econometrician) that both affect the firm long-term growth prospects and its probability to be financed by a VC. VC treatment could therefore be endogenous. In this case, the effect found when VC companies are compared with other startups, which have not been financed by VC, could be just the selection effect or a mix of selection and treatment effects.

The empirical literature on this topic, reviewed in Section 2, struggles more or less fiercely with this selection problem. Many papers use propensity score matching to obtain a sample of control firms that are similar to those financed but with regard to just some observable features. Few papers rely on IV strategies that also attempt to control for unobservable characteristics. Most of the papers focus on few output indicators. The evidence of important VC effects is stronger in the US experience than in Europe. The most frequent results are that VC investors tend to

 $<sup>^{3}</sup>$ The US Small Business Administration website reports that on average the exit happens 4 to 6 years after an initial investment; in Italy, AIFI, the Italian Association of Private Equity and Venture Capital, estimates an average holding period of 5 years.

<sup>&</sup>lt;sup>4</sup>The Fondo Italiano di Investimento SGR runs 2 VC funds of funds with a target funding of more than 200 million, partly covered by Cassa Depositi e Prestiti, a state-owned company; since 2012 to 2016 they invested more than 100 million in private Italian VC funds, whose size was around 400 million at the end of 2016. Invitalia Venture SGR, a subsidiary of a public agency, runs another fund with a target funding of 100 million, which should be reached also with the contribution of private investors: this is a fund that directly co-invests in innovative start-ups with other private operators. Their effect on the size of the Italian venture capital market is expected to be remarkable when considering that early-stage investments in Italy in the whole period 2012-2016 were a bit more than 400 million.

largely increase the size and the survivorship-rate of the firms they finance. Effects on other firms' characteristics, such as profitability, productivity, innovation, and, namely, financial structure and governance are sometimes documented, though less frequently analyzed, specifically altogether, due to the difficulty in gathering data.

One contribution of this paper is that we control for the selection effect by comparing VC firms with similar firms that have requested a VC financing, but were not able to get it by a *narrow* margin (late-stage discarded firms). First, considering in the control sample only firms that demand a VC intervention excludes self-selection bias and is an important control for firm *unobservable* features, mainly the desire and determination to grow, and therefore of firm growth perspectives. Secondly, since VC investment is not random, we select our control group by considering only firms that have been discarded *at the very last stage* of the screening process or in due diligence. This strategy is very similar to that applied by Greenstone, Hornbeck and Moretti (2010) in a very different framework.<sup>5</sup> The rationale is that by including in the control sample only the late-stage discarded firms we enhance the similarity with the sample of VC financed firms. The reason for which the deal has not been completed is likely not the quality of the project, but more reasonably the inability to find an agreement on the valuation of the idea, the lack of funds or of coordinated interest by different investors as deals are sometimes syndicated. The selection process has been very strict and meticulous: only 6 per cent of the sample of the initial applicants for VC funding are included in the control sample.

The similarity between financed firms and those of the control group supports our identification strategy. We verify that, *before VC financing*, the firms in the control group are very similar to VC backed firms as for almost all the observable characteristics available in our data, included the average credit score, a sort of proxy catching-up the whole risk and quality of the firm measured using balance sheets indicators. On this respect, we use many more variables than previous studies. Finally, the longitudinal nature of our data allows us to estimate a diff-in-diff model where we control for all unobservable firms' characteristics before VC financing. All in all, the differences that we find between treated and control sample firms *after* VC financing can hence be considered as a good measure of the VC treatment effect.

A second contribution of this paper is that we initially consider the population of the firms financed by private VC investors in Italy in the period 2004-2014 (293 startups), as reported in

<sup>&</sup>lt;sup>5</sup>The authors want to estimate the plant opening's spillover in the US and need to identify a county that is identical, in the determinants of incumbent plants' TFP, to that where the plant decided to locate. To this purpose, they use a ranking reporting the winner county as well as the one or two runner-up counties (i.e., the "losers") that have survived a long selection process, but *narrowly* lost the competition.

the Venture Capital Monitor by AIFI, the Italian Association of private equity and venture capital investors.<sup>6</sup> The AIFI dataset is one of the best representation of private VC investments in Italy; the dataset is not proprietary and can be used by other scholars to replicate the analysys: this is not frequent in VC studies often based on proprietary data (Kaplan and Lerner, 2016). After the merge with the Cerved dataset, from which we get firm balance sheets data, and imposing the essential condition that firms have data the year before VC financing, the number of VC backed firms decreases to 101. This sample is still representative of the initial population of VC backed firms according to industries and geographical areas.

Thirdly, unlike other empirical papers, we consider the VC effects on many different firm outputs: in detail, we evaluate the effects of VC investors on firm size, sales, profitability, credit score, financial structure, survivorship and innovation. We are specifically interested in the effects on firms' financial structure and loan terms in order to test whether VC investment creates a signaling effect for other investors, above all banks (Hellmann, Lindsey and Puri, 2008). Finally, we focus on understanding the main channels through which VC investors have a positive impact on the firms they finance by disentangling the pure financing effect, for equity provision, from the one due to VC management and network connections.

As a brief preview of the results, we find that VC investors have a fast and extended positive effect on the size of the firm: during the 4 years after VC financing, total assets increase on average by almost 800,000 euro more than that of firms that do not receive any VC finance (more than half of the average total assets before financing/rejection). Results on assets are confirmed by labour costs, mainly through an increase in the number of employees. A larger rise in labour costs with a similar trend in sales explains the worse profitability of VC-backed firms and the deterioration of their credit score: both these effects tend to disappear after 4 years from VC financing, when sales increases more for VC-backed firms, though with a large dispersion that makes not significant the difference with the control sample. We also uncover important effects of VC investors on innovation activity that develop 2/3 years after financing: both the probability and the number of patent applications increase more for VC-backed firms. No differences are detected for the survivorship rates.

As expected, equity increases much more for VC-backed firms (452,000 euro more with an average value of equity before financing of almost 400,000 euro); leverage consequently decreases. As for bank loans, we detect a larger shortening in debt maturity and a higher increase in the cost of debt

<sup>&</sup>lt;sup>6</sup>The analysis excludes corporate VC, i.e. VC investments made by non-financial corporations, and public VC that are not reported in the Venture Capital Monitor by AIFI.

for VC-backed firms, which are likely to be correlated with the worsening in their credit score. The effects on firms' size and innovation persist when the control sample is reduced to consider only rejected applicants that increased their capital. This means that VC positive effects on size and innovation are not only explained by equity financing: their managerial experience and networking connections play also an important role. Finally, the positive effects on size and innovation are exclusively driven by independent VC investors with respect to captive VC; the injection of equity of the former is much larger.

The plan of this paper is as follows. Section 2 reviews the empirical literature on this topic, while Section 3 explains our research desing based on the VC selection process of start-ups. Section 4 describes the data used and presents some descriptive statistics and Section 5 shows the main empirical strategy followed in the analysis. In Section 6 the results obtained when comparing VC treated firms with late-stage discarded firms are presented. In Section 7 some robustness and extensions of the analysis are considered, while Section 8 discusses the results and concludes.

### 2 Literature review

The empirical literature most related with this paper analyzes US companies. Most of the papers are aware of the selection problem, though only a few tackle it in a very comprehensive way by controlling for unobservable firm characteristics before VC financing. Helmann and Puri use a sample of Silicon Valley startups and do not control for other selection problems; they find that the startups receiving VC financing were faster in reaching the market with their products (Hellmann and Puri, 2000) and that venture capitalists also play an important role in the firm's organization, frequently replacing the founder with an outside CEO (Hellmann and Puri, 2002). Kortum and Lerner (2000) analyze the impact of VC on patents and they control for unobserved factors using a policy shift that freed pension funds to invest in VC in 1979 in the US; they find that increases in VC activity in an industry are associated with significantly higher patenting rates.

More recently, Puri and Zarutskie (2012) use a longitudinal dataset of private companies and match VC backed firms with others non VC backed firms using only size, sector, geographical area and age in the year that the VC financed firm receives the first round of VC; they find that VC financed firms achieve larger scale, but are not more profitable; default rates are also lower among VC backed firms. Chemmanur, Krishnan and Nandy (2011), using a very similar dataset but also different empirical strategies to control for unobservable firms' characteristics, find that VC backed firms have higher survivorship rate and total factor productivity, the main output they focus on. One of the most appealing studies as for the attempt to control for the selection problem is Kerr, Lerner and Schoar (2014). The authors compare firms financed by early stage investors (business angels in their case) with those that have been discarded with a level of score just below a threshold and that are hence very similar, in some observable and unobservable characteristics, to the firms that have been financed; they find that firms receiving financing by business angels have improved survival, exits, employment, patenting, Web traffic, and further equity financing.<sup>7</sup> Another study (Samila and Sorenson, 2011) points out some macroeconomic effects of an increase in the supply of venture capital, even when instrumented, in terms of firm starts, employment, and aggregate income.

Regarding Europe, the results about VC effects are weaker. Bottazzi and Da Rin (2002) develop a unique hand-collected data set recording the companies that went public on Euro.nm market from its inception in 1996 to December 2000<sup>8</sup>; they argue they consider only startups companies to reduce the bias in the comparison, similarly to what Hellmann and Puri (2000) did in the same period in the US. They find European venture capital to have a limited effect on firms' ability to grow, create jobs and raise equity capital; these results hold after matching firms using few observable characteristics. Weak results of VC on innovation are also found in Popov and Roosenboom (2012) who follow an approach similar to Kortum and Lerner (2000): they work with data on 21 European countries and 10 industries during the period 1991-2005 and use, as an exogenous variation of VC, data on fund-raising and on the structure of private equity funds in each country.<sup>910</sup> They find that VC investments seem to have an effect only in the sub-sample of high-VC countries and in countries with lower barriers to entrepreneurship, with a tax and regulatory environment that welcome venture capital investments, and with lower taxes on capital gains.

A couple of recent papers, mainly based on matching procedures and on the VICO dastaset<sup>11</sup>, find that independent VC have effects on sales growth and on exit performances of financed firms,

<sup>&</sup>lt;sup>7</sup>Another interesting paper regards China between 1998 and 2007: Guo and Jiang (2013) use a propensity score matching and also instrumental variable estimations based on the number of IPO in the stock market. They find that VC backed firms outperform non-VC backed in terms of profitability, labour productivity, sales growth, and R&D investments.

<sup>&</sup>lt;sup>8</sup>Euro.nm was the result of the alliance of Europe's new stock markets for innovative companies in high-growth industries along the lines of America's Nasdaq.

<sup>&</sup>lt;sup>9</sup>Their idea is that independent funds have to invest within a relatively short time window compared with captive funds that do not have a limited lifespan and do not raise capital from outside investors other than the single owner of the private equity fund (e.g. a bank or insurance company). Therefore, increased flows in venture capital translate into investments in companies at a faster pace when a country has a higher fraction of independent as opposed to captive VC funds.

<sup>&</sup>lt;sup>10</sup>They are able to replicate Kortum and Lerner (2000) results for the US in the same period; however, they also show that, even in the US, in a more recent period, VC had a comparably weak impact on innovation.

<sup>&</sup>lt;sup>11</sup>More than 8,000 European high-tech firms, of which less than 10 per cent are VC backed.

while no effects are detected for government-managed VC (Grilli and Murtinu, 2014; Cumming, Grilli and Murtinu, 2017). On a similar line of research, Bottazzi, Da Rin and Hellmann (2008), analyzing only VC firms, using a hand-collected sample of European venture capital deals<sup>12</sup>, find that investors' activism is more widespread among independent than captive VC (bank-, corporateor government owned) and is positively related to the success of portfolio companies which is measured with a successful VC exit, either through an IPO or an acquisition. However, they do not look at specific different outputs of financed companies and they do not compare VC backed firms with others.

As for Italy, some empirical papers use a dataset built by the Politecnico of Milan, based on a sample of high-tech startups followed between 1993 and 2003, of which around 10 per cent were VC backed.<sup>13</sup> One of the most interesting result is obtained in Bertoni, Colombo and Grilli (2011): after controlling for selection of the unobservable variables with a panel fixed effect estimation, this study finds that VC financing spurs firm growth.

## 3 Selection process among venture capitalists and research design

How do VC investors decide whether or not to finance an innovative startup?

A typical flow chart is reported in Figure 2. VC investors receive thousands of requests of financing each year. Normally the entrepreneurs send a copy of their business plan or an executive summary. Most of them (50 per cent) are rejected after an initial and rapid evaluation of the papers. A share of startups of around 20 per cent reach the phase of a deeper evaluation. At this stage VC investors meet the team and conduct a broad analysis of the data; the startup team is invited to give a short presentation, which is followed by a question-and-answer session. They also analyse the business plan, the way the idea can be protected, the team experience in the market and its commitment in terms of time and funds devoted to the development of the idea, commercial and/or industrial partnerships. For the most promising ideas, VC investors also start to think about the structure of the operation, i.e. the terms of VC entrance and exit and the valuation of the firm. The most promising companies arising the greatest interest (around 10 per cent) enter a costly due diligence process during which the structure of the operation is finalised. Eventually, only 2-3 per cent of the ideas are financed.

There are many reasons why a deal is not reached during the screening or in due diligence. Most

<sup>&</sup>lt;sup>12</sup>They analyze 1652 companies financed in 17 European countries by 119 venture capitalists between 1998-2001.

<sup>&</sup>lt;sup>13</sup>The same dataset is included in the VICO dataset at the European level, mentioned in the previous paragraph.

of them arise quickly in the process and are related to the quality of the firm, i.e. an inadequate business plan, an idea that is not developed enough, poor quality and/or low commitment of the management team. Some of the reasons are not related to the quality of the idea, but arise from VC preferences for some industries, in which they are specialized, or for the envisaged size of the business that could often be considered too small or too large.

Some other reasons might arise later in the process and are mostly related to the lacking agreement on the terms of entrance and exit of the VC fund in the firm and its evaluation, or to the difficulties in finding co-investors in the deal, or the absence of an envisaged way out. Trust is also quite important: early-stage investors take on significant risks as there are often many unknown factors. VC must be confident that the management team will be able to adapt to new conditions without losing focus. VC investors can hence change their mind about a startup also in the final stage of screening or even in due diligence. Moreover, trust is a matter of chemistry, not necessarily connected with the quality of the business. It could be that some startups are rejected by a VC for lacking of trust, but the overall idea is good.

Our research design is based on singling out the *late-stage rejected business* in the idea that at this stage the reasons for which the deal has not been concluded might be those mentioned in the previous paragraph and are less likely to be related to the quality of the idea or of the management. All in all, we try to select the best projects among those that have been rejected.

We are able to build this control sample as we have information on a sub-sample of startups rejected at the different steps of the screening or during the due diligence process. We asked all the VC members of the AIFI to share with us confidential information about the companies that applied for venture capital and their subsequent evaluations. Five of them, which account for one fourth of all the investments undertaken in the period 2004-2014, gave us the information we need. We thus know the tax code of more than 4000 companies that applied for this source of financing during the period 2006-2014, the year in which the screening process occurred, and the stage of the process when the applicant has been rejected. Albeit these VC use different ways of ranking firms, we were able to single out for each investor those businesses that were discarded *at the very last stage* of the screening process or in due diligence and with the highest grades.<sup>14</sup>

In conclusion, this research design allows us to identify the best applicants that were not able to get VC financing. We end up with 258 firms in the control group that account for almost 6

<sup>&</sup>lt;sup>14</sup>Some VC gave a summary grade to the applicant, others comment about the reasons not to undertake the investment. For some VC we choose discarded firms among those with the highest grades, for others the descriptions and comments given by the investors implied they were among the best of rejected firms.

per cent of all VC applicants for which we have information, a percentage that is very similar to the difference between the share of firms reaching the last step of the evaluation and that of firms financed by VC (Figure 2).

### 4 Source of data and descriptive statistics

Aside from information on rejected startups among a sub-sample of VC applicants, the analysis is based on data coming from three other different sources. The first source of data is the annual survey Venture Capital Monitor by AIFI. We use the surveys between 2004 and 2014 to identify the universe of venture capital deals in the period (293 VC investments). For each deal we observe the name and the origin of the target firm, the name and the type of the investors, and, for most of the investments, some other details, such as the amount invested and the share of the firm acquired by VC. More specifically, about three-fourths of the target companies are private enterprises, 9 per cent are corporate spin offs and 15 per cent university spin offs. There are 82 different investors: many of them are however associated to only one deal, whereas the most active venture capital has invested in 17 different firms. As for deal terms, the amount invested is specified for more than 70 per cent of the investments: the average and the median value of the investments are 2.5 and 1 million of euro, with a range from 0.1 to 66 millions; 30 per cent of the deals are syndicated. Regarding the years of investments, prior the financial crises the trend in total number of deals was increasing, a pattern that has recovered starting from 2011.

Secondly, for every company in our study we gather information for the period 2000-2015 using the Cerved database that contains detailed annual balance sheets for all limited liability companies based in Italy. In the analysis we only focus on active firms with available information at least one year before the VC treatment. This condition reduces the number of ventures in our study to 101, but is crucial to evaluate the level and trends of the variables of interest since the year before the treatment. In order to evaluate the representativeness of this smaller sample of the initial population of VC-backed firms, in Table 1 we compare their industry and geographical distributions that turn out to be very similar, while firms in our sample tend to be slightly more innovative when considering the probability and the number of patent applications.<sup>15</sup>

As mentioned, we focus on different firm characteristics, such as size, profitability, and financial

<sup>&</sup>lt;sup>15</sup>Industry and geographical area are available in the the Venture Capital Monitor, while patent applications are found in the Orbis database as explained in the final paragraph of this section. The status of limited liability company, determined by the use of the Cerved database for the balance sheet data, is likely to have low impact on the representativeness of our sample as the innovative start-ups included in the register since the 2012 Law, which has created them and given them important fiscal benefitis, need to have this legal form.

structure. As for size, we present results on total assets, labor costs and sales; we are able to use the number and wages of employees by incrementing the Cerved database with data from INPS (the Italian retirement management agency). Our measures of profitability are EBITDA/Assets and ROE, whereas for financial structure we focus on book value of equity, total financial debts, and leverage, which is defined as the ratio between financial debts and the sum of equity and total financial debts. Moreover, in order to capture the relationship with banks, we also consider the ratio between bank debt and total financial debts, the ratio between short term bank debt and total bank debt, and the cost of loans.

Our group of VC-backed firms is therefore composed by 101 ventures financed over the period 2004-2014 for which we have balance-sheet information in the year before the treatment. Table 2 provides summary statistics on these VC-backed firms: 58 per cent of them are located in the North of Italy, whereas 24 per cent operate in the Center and 19 per cent in the South. About 70 per cent of these companies operate in sectors with high-growth potential, that is ICT, telecommunication, engineering, and pharmaceuticals, 17 per cent of them work either in the energy sector or in manufacturing, whereas 14 per cent in other services. As expected, these firms are young (5 years on average), small, as the size dummy, which reflects different accounting variables such as assets and labour costs and whose range is between 1 and 4, is on average equal to 1.1, and have a large incidence (71 per cent) of intangible assets on total fixed assets (tangible and intangible assets). They are also not profitable and, much less expected, their leverage is high (96.6 per cent), tough three quarters of their bank loans have a maturity shorter than 1 year. According to the score provided by Cerved, they are quite risky firms. The score in Cerved is particularly important as it captures the intrinsic quality of a company: the average rating for the treated firms is 6.5 out of 9, where higher values mean higher risk.<sup>16</sup>

Finally, as a measure of innovation we collect patent applications from the European Patent Register, which is kept by the European Patent Office, such as reported in the Orbis database. We focus on patent applications, rather than grants, to conform with most of the empirical literature about innovation. Using this dataset, we augment the Cerved dataset on balance sheets with information about the total number of patent applications at the European Patent Office by each firm in every year.<sup>17</sup>

 $<sup>^{16}</sup>$ Cerved calculates the Z-score on the basis of different balance-sheet indicators and assigns firms in different 9 risk classes, from safe (1-4), to vulnerable (5-6) and risky (7-9).

<sup>&</sup>lt;sup>17</sup>As in three out of four sources of data, firms' identifiers are names rather than fiscal codes, we double check that merges with the Cerved dataset are correct using the Business Register kept by the Italian Chambers of Commerce (https: //telemaco.infocamere.it.).

## 5 Empirical strategy

To assess whether firms that benefitted from VC financing afterward outperform those that did not receive VC funding is a challenging task, as mentioned in the Introduction. In order to identify the impact of VC financing, recipient and non-recipient firms should differ only for the assignment of the funds. This assumption is not easily testable and could be affected by two sources of bias that we need to address in order to correctly identify the impact of VC financing.

The first source of bias comes from firms' self-selection. Enterprises that apply for VC funding can be different from those that do not. The decision to apply may be related to the quality of the new idea and the willingness to economically exploit it, or to other unobservable characteristics of the firms that are correlated with the firm performance. In these circumstances, comparing the results of recipients with those of non-recipient firms that do not apply for VC funds might produce biased estimates of the effects of the VC financing.

The second source of bias is due to the non-random assignment of VC. Recipient firms might be inherently different from those that applied, but were not financed. VC investors could select the best high-tech startups, and unobservable firm features might affect both the firm probability to be financed by a VC and its long-term growth prospects. Again, this type of problem induces a bias in the estimation of the effect of the financing to the extent that firm characteristics for which we are not able to control for are correlated with the firm performance and differ between recipient and non-recipient firms. To deal with these issues, we use an identification strategy based on a careful selection of the control group and diff-in-diffs estimation method.

The availability of the information on rejected applicant firms allows us to fully control for the first source of bias, i.e. self-selection. We use rejected applicants as the set of firms from which we choose the control group for financed firms. Since both groups of firms self-select among the applicants they cannot differ in this respect; hence self-selection bias does not occur.

Our strategy tries to control as much as possible also for the second source of bias. As carefully explained in Section 3, we exploit the multi-step screening process of VC investors and the grades they assign to the applicants to build a control sample of firms that were rejected in the final stages of the screening process or in due diligence.

To evaluate the validity of our identification strategy, we carefully verify whether VC backed firms and those in the control group are very similar before VC financing in terms of a larger set of observable characteristics than that used in previous studies. We consider indicators of size, profitability, financial structure, innovation and some other variables including a synthetic measure of the risk of corporate failure (Z-score calculated by Cerved), which is very useful as it is an index of the overall quality of the firm, able to catch-up some unobservable firm characteristics such as for example the ability of the firms' management team. The results are very clear-cut. Even without imposing any matching, there are no statistically significant differences between VC-backed and late stage rejected firms (Table 3), but for the initial age of the firm that we hence include in our estimations as control.

Finally, in order to control for any residual differences in unobservable firm characteristics between financed and rejected firms *before* VC financing, we exploit the longitudinal nature of our data and use the diff-in-diffs (DID) estimation method. Using the DID, the effect of the VC financing is estimated by the change in the difference of the output between recipient and non-recipient firms before and after the VC investment.

Formally,

$$DID = [E(Y_{it*+x}^1) - E(Y_{it*+x}^0)] - [E(Y_{it*-1}^1) - E(Y_{it*-1}^0)]$$
(1)

where E is the average value,  $Y_i$  is the outcome variable of the firm i, t<sup>\*</sup> is the year of VC financing, x are the number of years after VC financing (1 to 4 years) and the top index 1(0) refers to the VC-backed firms (control firms).

The DID method is strongly dependent on the parallel trend assumption, i.e. is based on the assumption that without the VC financing the outcome variables of the two groups would have followed the same time paths. Therefore, we carefully verify this hypothesis by testing the similarity of outcome variable trends in our samples before the treatment. The results are plotted in Figures 3 and 4: they indicate very similar trends before the VC financing for the main outcome variables analyzed in the paper. These graphs are also very useful because they show graphically the effects of VC financing on selected firms' outputs.

In detail, our baseline model is:

$$y_{it} = \beta_1 * post_t + \beta_2 * VC_i + \beta_3 * post_t * VC_i + \beta_4 * dyears + \beta_5 * f_i + const + \epsilon_{it}$$
(2)

where  $y_{it}$  are the outcome variables (assets, sales, labor costs, etc.), *i* is an index for firms, *t* refers to different years,  $VC_i$  is a dummy equal to 1 for firms that are financed by VC investors, *dyears* are year dummies to control for different economic cycles and  $f_i$  stands for the firm fixed effect to control for unobservable firm characteristics that are fixed over time; in this equation  $beta_4$  and  $beta_5$  are vectors of coefficients. Standard errors are clustered at firm level to take into account

the correlation among the observations of the same firm.

As for the term  $post_t$ , first we run a DID estimation collapsing the various  $post_t$  terms in a single dummy post to capture the overall effect of VC financing since the year of financing/rejection over the 4 years afterwords. Then we run 5 different DID estimations with the variable  $post_t$  (t = 0, ..., 4) defined as dummies taking values 1 the year of financing/rejection or one of the 4 years afterwords, 0 the year before financing/rejection and missing otherwise: in this way we study the effects of VC year by year. In other words the dummy  $post_t$  is equal 1 in the year when we want to evaluate the VC effect on the firm, 0 in the year before financing/rejection and missing otherwise.<sup>18</sup> The parameter of interest is  $beta_3$ , that of the interaction term  $post_t * VC_i$ , which is reported in the tables.

One potential drawback of DID estimates is that they could be biased if the outcome variable of VC financed and VC non-financed firms have different trends. Apparently, from the figures this does not emerge. In any case, we control also for potential differences in time trends by interacting some pre-financing control variables, such as the initial age of the firm at financing/rejection, geographical area and sector dummies, with the post financing dummies  $post_t$ , in the idea that firms in different steps of life-cycle, belonging to different sectors or geographic areas could be subject to different time trends. In a less parsimonous specification of the previous estimation we hence include also the following control variables, where all coefficients stand for vectors of coefficients:

 $\beta_6 * init.age_i + \beta_7 * init.age_i * post_t + \beta_8 * sec_i + \beta_9 * sec_i * post_t + \beta_{10} * area_i + \beta_{11} * area_i * post_t$ (3)

# 6 Results of the effects of VC financing

In this section we present the results concerning VC effects on firm's size, activity, innovation and financial structure. From Figures 3 and 4, in which we include graphs for selected variables that show some changes between VC backed and non-VC backed firms, the evidence is that after the VC intervention we observe a much stronger increase in total assets and labour costs over the entire period of the analysis. There is also a positive effect on firm sales, though only after 4 years from VC financing. We also observe a negative trend in the firm profitability (EBITDA/total assets) for VC backed firms, which also vanishes after 4 years since the VC financing, consistently with the

<sup>&</sup>lt;sup>18</sup>In order to avoid Betrand, Duflo and Mullainathan (2004) criticism, the estimations by years include the period -1 and separately each single year in the post financing period, thus only two periods are included in each estimation: -1 and 0; -1 and 1; -1 and 2; -1 and 3; -1 and 4. Alternatively we present also the results of the estimations over the whole post financing period taking the average of each variables between 0 and 4 over the post period.

surge in sales. The figures also show that VC backed firms tend to have a much higher equity<sup>19</sup>, more innovation activity and lower survival rate.

We then verify the previous graphical evidence in a multivariate econometric setup. In Tables 4 to 6 we report the results for the coefficient  $beta_3$  of DID estimations (equation 2). We run the estimations with no controls and with all controls, including initial age, area and sector and their interactions with the term  $post_t$ . As the results of the two specifications are similar, we report in the tables only those obtained with all controls. Most of the graphical evidence is confirmed. In the tables we show first the effect on the whole period since VC financing and then the one for each single year.

First, we find that VC investors have a rapid and extended effect on firms' size: during the 4 years after the VC financing, total assets increase on average by 780,000 euro more for VC backed firms than for firms not receiving any VC financing (Table 4) a bit more than half of the average total assets of companies before financing. This is the average effect on firm size over the 4 years after the VC financing; from the interaction dummies, which capture the trends year by year, we elicit that the effect on firm size is increasing over time: after 4 years from VC financing the increase in assets is almost 2 million of euro more than for the control group. The gradual increase in firm size is confirmed by the rise in labor costs: on average roughly a rise of 157,000 euro more for VC backed firms with respect to an average amount of labor costs before VC financing equal to 280,000. Furthermore, the last two columns show that the increase in labor costs is due almost exclusively to a rise in the number of employees (increasing by 2 units more for VC backed firms), while the difference in the increase of monthly wage is positive but not significant.

As for sales, the effect of VC is increasingly positive, though never significant due to the large heterogeneity in the results which reflects in high standard errors. This could be a consequence of projects financed by VC that frequently take more time to reach the commercialization phase, i.e. projects that are in an earlier stage of their life-cycle and hence riskier. As a consequence of the gradual upsurge in sales, the operating profitability (EBITDA/assets) of firms that got VC financing, which was initially much worse than that of control group, improved; after 4 years from financing the difference between VC backed firms and control sample is no longer significant (Table 5. Moreover, there are almost no differences in the return on equity (ROE) of the two groups of firms. Nonetheless, the strongest negative trends in operating profitability for VC-backed firms is

<sup>&</sup>lt;sup>19</sup>Rejected applicants do not get any equity financing from VC operators, but they might get equity from other investors. Indeed, investors in the capital of innovative start-ups, like friends, small entrepreneurs and corporates, have benefitted from fiscal incentives introduced with a Law passed in 2012.

likely to explain their worse rating, measured by an increase in the Z-score index by 0.6 points more than that of non-VC backed firms (the average score before financing is 6.5); consistently with the improvement in operating profitability, this difference vanishes after 4 years.

We then focus on firm financial structure indicators that are seldom analyzed in previous studies (Table 6). We find a remarkable stronger increase in equity for VC-backed firms: 452,000 euro higher than for the control group, more than double the average equity of firms before financing. The increase in equity becomes more and more wider, suggesting a multi-stage process of financing. This considerably reduces more the leverage of VC-backed firms (64 percentage points of additional reduction compared with a leverage before financing of VC-backed firms of 96.6 per cent). Overall VC-backed firms have a much more capitalized and hence stronger financial structure after VC financing. It is worth noticing that the additional increase in total assets for VC-backed company is much larger (almost twice as much) than that in equity: there is therefore a multiplicative effect induced by VC activity; we will deepen more thoroughly this issue in the next section.

Financial debts of VC financed firms also increase more than for the control sample, though the high variance of the results makes the difference not significant. Interestingly, VC-backed firms tend to have a shorter debt maturity than firms in the control sample (an increase of 10.6 percentage points more in the short term debt share compared with an average of 75 per cent before financing) and pay a higher interest rate on their financial debt (an increase of 7 percentage points more than for the control sample, compared with an average cost of funds of 4.5 per cent before the treatment). These worse conditions on bank loans might be explained by the deterioration in operating profitability and credit score; this seems specifically true for the cost of funds for which the differences tend to disappear after 3 years since VC financing when the differences in score also vanish.

We finally deepen the evaluation on innovation activity and survival rates using the DID estimations (Table 5). When considering a dummy equal to 1 for firms that applied for a patent, the estimations show that the effect of VC financing on the whole period is positive, but not statistically significant. However, the increase in the cumulated number of patent applications is much larger for VC backed firms: a rise of 0.25 more patent applications than for the control sample, almost twice as much as the average number of patent applications before financing/rejection. When analysed over time, the effects on firms' innovation develop clearly 3-4 years after financing; this is expected as it takes time to strengthen an idea to the point of asking for a patent: after 4 years of financing, VC-backed firms show a much higher increase in patent applications (1.6 more) compared with the control sample. We have reported in the table the results obtained with linear estimations that allow us to use the same controls as for the other output indicators, including the firm fixed effect; we also verify the evidence regarding innovation with non-linear estimations such as probability and negative binomial models. Finally, we do not detect any significant difference in the firm survivorship rate after three years of VC financing or rejection.

## 7 Robustness and extensions of the results

#### 7.1 Comparison with firms in the control sample that increased their equity

In this subsection we show the results of some estimations regarding a control sample of laterejected firms that also increased equity thanks to investors different from venture capitalists. The main intent of this exercise is to evaluate whether the VC effects on firms' size and innovation are exclusively connected with equity financing or there are some effects linked to their managerial expertise or networking connection. Results are reported in Tables 7 to 9 and refer to a control sample of 163 firms compared with an initial initial control sample made of 258 firms.

The evidence is that even restricting the control sample in this way, the effects of VC financing on firm size and innovation are very similar to those presented in the previous section; this is also true for the results concerning the worsening of profitability and credit score (Tables 7 and 8). It seems therefore that the VC effects on firms' growth and innovation are related to the general activity of venture capitalists, and not only to the fact that they offset a funding gap with equity financing.

It is however important to underline the fact that even restricting the control sample to rejected firms that also got some equity financing from outside investors, the increase in equity for VCbacked firms is much stronger, similarly to what we have shown in the previous section (an increase in equity of 448,000 more for VC-backed firms; table 9). It is therefore possible that some rejected firms get equity from other investors, but the amount they gather is so tiny that the previous conclusion appears not well grounded.

We therefore further restrict the sample to rejected firms that rise equity and for which this increase is higher than a certain threshold (the 1st quartile of the distribution of the increase in equity). In this case the rise in equity for VC-backed firms is not significantly different than the one observed in this much smaller control sample (122 firms), while all previous results on the size, innovation and activity of firms financed by VC investors are confirmed.<sup>20</sup>

<sup>&</sup>lt;sup>20</sup>To preserve space results are not reported; they are available upon request.

The overall take of this extension of the analysis is hence that VC effects on firms' size and innovation of the firms they finance are not only mechanically linked to their equity financing.

#### 7.2 Captive and independent venture capitalists

Another important issue is whether there are differences in the effects of the firms that have been financed by captive VC (bank-, financial or insurance company-owned in our sample) and independent VC investors. Captive VC do not raise capital from outside investors other than the single owner of the private equity fund and they could have specific indications, from the single owner, about the investment policy to adopt. Independent VC investors gather funds from the market and they are freer to chose the companies in which to invest. In our sample of 101 startups, 42 have been financed by captive VC and the remaining 59 by independent VC.

In order to test the differential effects of the two categories of VC, we split the crucial interaction term - post\*VC - in the equation 2 using two dummies for VC: the first referring to captive VC and the second to independent VC. For each period, we report the coefficients of two interaction terms - post\*VC-captive and post\*VC-independent - measuring the effect of each specific group of VC after their financing.<sup>21</sup>

In Table 10 the evidence is that the growth in total assets for the whole period after VC financing is stronger, compared with the control group, only for independent investors. Similar results hold for labors costs and the number of employees that increase more, compared with the control group, only for VC-backed firms financed by independent operators. All in all, the positive effect of VC on the size of firms arise only when financing is obtained by independent investors. Similarly, for innovation activity in Table 11 the evidence is that the positive effect on the number of patent applications after 3 years is entirely driven by indipendent VC investors, while some effects on the probability of patent applications are detectable also for captive VC investors.

The previous findings are strictly connected to what we observe in the financial structure of the firms. Equity increases much more for VC backed firms than for firms in the control sample, but only when they are financed by independent VC (Table 12). When the firm is financed by a captive VC, its equity has the same path as for the firms in the control group one year after financing, suggesting that the injection of capital is much smaller and limited in time. Conseguently only firms financed by independent VC investors show a much stronger reduction in leverage compared with the one observed in the control group.

<sup>&</sup>lt;sup>21</sup>The dummy VC has been similarly split in two dummies.

On the contrary, there are no remarkable difference as for operating profitability and credit score that are worse for all VC-backed firms, regardless of the type of VC investors (Table 11). However, the worsening of credit score and operating profitability has different effects on banking loan terms: as for firms financed by independent VC, interest rates increase much more than for the control group (almost 10 percentage points more), while for firms financed by captive VC we observe a much stronger increase in the share of short-term bank loans (16 percentage points more; Table 12).

All in all, independent and captive VC investors appear to be characterized by very different investment attitudes. Italian independent VC show greater activism in line with what has been found in other European countries (Bottazzi et al., 2008; Grilli and Murtinu, 2014; Cumming et al., 2017). They invest important amount of capital in the firms held in their portfolio that consequently grow faster and innovate more. On the contrary, captive VC invest less money in startups that are hence gathering an amount of equity similar to those obtained by firms in the control sample; for these investors we detect no effect in term of faster growh of the firms and much weaker effects as for innovation.

#### 8 Discussion of results and conclusions

In this paper we use a novel strategy to tackle the selection problem influencing all the evaluation exercises of VC activity. On the one hand, we get rid of firms self-selection by considering in the control sample firms that have also looked for VC finance. On the other hand, we deal with the selection made by VC investors considering only late-stage discarded firms in the idea that firms that narrowly lost the contest were more similar to financed firms. This strategy is very similar to the one used by Greenstone et al. (2010) when tackling a very different empirical issue.

Although starting with the whole population of firms financed by private VC investors in Italy in the period 2004-2014, as reported in the Venture Capital Monitor by AIFI, when we impose the essential condition that firms have a balance sheet one year before VC financing, we end up with a sample of firms equal to one third of the universe; though we assess the representativeness of our sample in terms of geographical areas and sectors, it is true that the results can be generalized only with caution.

The evidence is that VC investors are able to accelerate the growth of the firms they finance and help their innovation activity. These firms show a larger increase in size (total assets, labor costs, no. of employees) and they innovate more (in term of the probability and number of patent applications) compared with very similar firms in the control sample. This is not just a mechanical effect of the injection of equity capital. First, we notice that the larger increase in assets for VC-backed firms is by far greater than the wider rise in their equity. Secondly, we repeat the exercise by considering only firms in the control sample that also increase equity thanks to other investors (family, friends, corporate, etc) and the results still hold. The positive effects of VC investors in terms of firms' growth and innovation are hence likely to be connected also with their managerial expertise or network connections.

In general, an unexpected result is that all the innovative startups analyzed have a high leverage in the year before VC financing or rejection. This is actually in line with was has been discovered in the US by Robb and Robinson (2012), who find that new firms, even the home-based ones, analyzed for the period 2004-2007 rely heavily on external debt sources, such as bank financing: when summing up all forms of debt, it accounts for more than 50 per cent of the total capital of the firm. Similar recent evidence is found for Italy (Bonaccorsi di Patti and Nigro, 2017). Still, we focus in this paper on innovative startups, which are riskier and with a high share of intangible assets, for which bank lending is not the more appropriate source of finance. Consistently, Brown, Fazzari and Petersen (2009) find that for the US high-tech listed firms the share of new net debt issues on total net finance is very low, less than 2 per cent and that of net equity is higher (29 per cent); corresponding figures for Italian high-tech listed firms were reversed for the period 1998-2006 (Magri, 2014). In this paper, the evidence for a more recent period (2004-2014) is indeed that for VC-backed firms the wider increase in equity also mirrors in a stronger financial structure after VC financing: their leverage hence decreases much more than for firms in the control sample.

As for the effects on other sources of finance different from equity, we find that financial debts increase more for VC-backed firms though there is large heterogeneity: the differences are hence not significant. It is likely that the higher banks' selectivity after the 2008 financial crisis had an impact on these results given that VC-backed firms are quite risky firms. Due to the very innovative nature of their ideas, which delays the commercialization of products and services, and the upsurge in labor costs, their operating profitability is much worse than that of non treated firms. This mirrors in a worsening in credit score for VC-backed firms that is likely to be the culprit of the larger increase in interest rates and in the share of short-term bank loans that we observe for them.

Finally, the positive VC effects on faster growth and innovation are exclusively driven by independent VC investors. Firms financed by captive VC investors (bank-, financial or insurance companyowned in our sample) have the same growth in size, equity and patent applications that those in the control sample. This evidence is line with some recent literature that shows more activism and results for independent VC investors (Grilli and Murtinu, 2014; Cumming et al., 2017; Bottazzi et al., 2008). Specifically, independent VC investors finance their firms in subsequent stages and this is likely to help them as it takes time to reach the point where a patent could be asked for. To support firms' innovative ideas and their profitability and growth, a longer period of time and patience is likely to be required (Mazzuccato, 2013).

# References

- Bertoni, F., Colombo, M. and Grilli, L. (2011). Venture capital financing and the growth of hightech start-ups: disentangling treatment from selection effects, *Research Policy* 40: 1028–1043.
- Betrand, M., Duflo, E. and Mullainathan, S. (2004). How much should we trust differences-indifferences estimates?, *The Quarterly Journal of Economics* **119**(1): 249–275.
- Bonaccorsi di Patti, E. and Nigro, V. (2017). The financing of start-up new firms in Italy: a first exploration, *Occasional Papers Forthcoming*, Bank of Italy.
- Bottazzi, L. and Da Rin, M. (2002). Venture capital in Europe and the financing of innovative companies, *Economic Policy* **17**: 229–270.
- Bottazzi, L., Da Rin, M. and Hellmann, T. (2008). Who are the active investors? Evidence from venture capital, *Journal of Financial Economics* **89**: 488–512.
- Brown, J., Fazzari, S. and Petersen, B. (2009). Financing innovation and growth: cash flow, external equity, and the 1990s R&D boom, *The Journal of Finance* **64**(1): 161–85.
- Chemmanur, T., Krishnan, K. and Nandy, D. (2011). How does venture capital financing improve efficiency in private firms? A look beneath the surface, *The Review of Financial Studies* 24(12): 4037–4090.
- Cumming, D., Grilli, L. and Murtinu, S. (2017). Governmental and independent venture capital investments in Europe: a firm-level performance analysis, *Journal of Corporate Finance* 42: 439–59.
- Gompers, P. and Lerner, J. (2001). The venture capital revolution, *Journal of Economic Perspectives* **15**(2): 145–168.
- Greenstone, M., Hornbeck, R. and Moretti, E. (2010). Identifying agglomeration spillovers: evidence from winners and losers of large plant opening, *Journal of Political Economy* **118**(3): 536–598.
- Grilli, L. and Murtinu, S. (2014). Government, venture capital and the growth of european high-tech entrepreneurial firms, *Research Policy* **43**: 1523–1543.
- Guo, D. and Jiang, K. (2013). Venture capital investment and the performance of entrepreneurial firms: evidence from China, *Journal of Corporate Finance* 22: 375–95.

- Hellmann, T., Lindsey, L. and Puri, M. (2008). Building relationships early: banks in venture capital, *The Review of Financial Studies* **21**(2): 513–541.
- Hellmann, T. and Puri, M. (2000). The interaction between product market and financing strategy: the role of venture capital, *The Review of Financial Studies* 13(4): 959–984.
- Hellmann, T. and Puri, M. (2002). Venture capital and the professionalization of start-up firms: empirical evidence, *The Journal of Finance* **57**(1): 169–197.
- Kaplan, N. and Lerner, J. (2016). Venture capital data: opportunities and challenges, Working Paper No.22500, NBER.
- Kerr, W., Lerner, J. and Schoar, A. (2014). The consequences of entrepreneurial finance: evidence from angel financings, *Review of Financial Studies* 27(1): 20–55.
- Kortum, S. and Lerner, J. (2000). Assessing the contribution of venture capital to innovation, RAND Journal of Economics 31(4): 674–692.
- Magri, S. (2014). Does issuing equities help R&D activity? Evidence from unlisted Italian high-tech manufacturing firms, *Economics of Innovation and New Technology* 23(8): 825–854.
- Mazzuccato, M. (2013). The Entrepreneurial State. Debunking Public vs Private Sector Myths, Anthem Press, Wimbledon.
- Popov, A. and Roosenboom, P. (2012). Venture capital and patented innovation: evidence from Europe, *Economic Policy* **27**(71): 447–482.
- Puri, M. and Zarutskie, R. (2012). On the life cycle dynamics of venture-capital and non-venturecapital financed firms, *The Journal of Finance* 67(6): 2247–2293.
- Robb, A. and Robinson, D. (2012). The capital structure decisisons of new firms, *Review of Financial Studies* 1(1): 1–27.
- Samila, S. and Sorenson, O. (2011). Venture capital, entrepreneurship, and economic growth, The Review of Economics and Statistics 93(1): 338–49.

Tables and figures

Figure 1: Venture capital investments as a percentage of GDP



Source: AIFI for Italy, AFIC for France, EVCA-BVKA for Germany, ASCRI for Spain and NVCA for the United States.

Figure 2: Selection process among venture capitalists



# Table 1: Comparision between samples of VC-backed firms

Percentage values, frequency, and numbers

	Our sample	VC population		Our sample	VC population
Sector			Geographical area		
Business services	3	1	North-west	40	49
Clean tech	7	6	North-east	18	13
Construction	2	2	Center	24	20
Consumer goods	3	2	South-islands	19	17
Financial services	2	2	Innovation		
Food and beverages	1	2	Probability patent application	0.3	0.2
Health care and social services	5	5	No. patent applications	0.12	0.08
ICT	36	37			
Industrial products	9	9			
Leisure	1	1			
Media and communications	7	7			
Nanotech	2	1			
Other professional and social services	6	6			
Pharmaceutical and biopharmaceuticals	11	13			
Transportation	1	1			
Utilities	2	3			
Web and mobile applications	3	2			
Total	100	100	Total	100	100
N	101	293		101	293



Figure 3: Trends in some output variables: late stage rejected control sample



Figure 4: Trends in some output variables: late stage rejected control sample

(c) Number of patent applications

(d) Survivorship

		Summary Statistics VC		
Area			Size	
North-west	40	40%	Total Assets (*1000 euro)	1506
North-east	18	18%	Size dummy	1.1
Center	24	24%	Labor costs (*1000 euro)	281
South-islands	19	19%	Sales (*1000 euro)	814
Year of Financing			Profitability	
2004	3	3%	EBITDA/Assets $\%$	-11.5
2005	3	3%	ROE %	-59.1
2006	6	6%	Financial structure	
2007	8	8%	Leverage %	96.6
2008	6	6%	Short-term bank debt/Bank debt $\%$	75.2
2009	4	4%	Equity/Assets %	20.2
2010	7	7%	Financial costs/Financial debts $\%$	4.5
2011	19	19%	Innovation	
2012	6	6%	Probability patent application	0.12
2013	17	17%	No. patent applications	0.35
2014	22	22%	Other characteristics	
Sector $\%$			Age (years)	4.9
Manufacturing	8	8%	Intangible assets/Tangible+Intangible assets $\%$	71.3
Energy	9	9%	Rating	6.5
IT	39	39%		
Telecomunication	5	5%		
Engineering	6	6%		
Pharmaceutics	20	20%		
Other services	14	14%		
N	101			

## Table 2: Summary statistics for venture-backed firms

The statistics for area, sector, size, profitability, financial structure and other characteristics are calculated in the year before treatment.

Table 3	B: Ba	lancing	properties	between	treated	and	$\operatorname{control}$	groups
---------	-------	---------	------------	---------	---------	-----	--------------------------	--------

	VC-backed(1)	Late stage rejected $(2)$	t test (2)-(1)
Size			
Total Assets(*1000 euro)	1506	1648	0.44
Labor Costs (*1000 euro)	281	298	0.20
Sales(*1000 euro)	814	1242	1.42
Profitability			
EBITDA/Assets %	-11.5	-2.5	1.62
ROE %	-59.1	-63.2	-0.06
Financial Structure			
Leverage %	96.6	58.5	-1.68
Financial debts(*1000 euro)	544	576	0.20
Equity(*1000 euro)	393	324	-0.60
Bank debts/Financial debts %	57.5	60.2	0.48
Short-term bank debts/Bank debts $\%$	75.2	79.4	0.83
Financial Costs/Financial debts $\%$	4.5	6.3	1.06
Innovation			
Probability of patent applications	0.12	0.14	0.42
No. patent applications	0.35	0.34	-0.037
Other characteristiscs			
Age	4.9	7.5	2.5
Rating	6.5	6.2	-1.46
N	101	258	

Table 4: Effects of venture capitalists on firms' size and activity indicators Diff-in-diff estimations (coefficient  $beta_3$  is reported) - different post-treatment periods

Post-treatment periods	Assets (*1000 euro)	Labor costs (*1000 euro)	Sales (*1000 euro)	Employees number	Monthly wage euro
average post-treatment	780.4 (378.2)**	157.3 (42.5)***	126 (163.3)	2.0 (1.2)*	189.6 (191)
$t^*$ (year of financing)	607.2 (333.4)*	89.9 (26.0)***	-29.8 (132.1)	4.2 (1.6)***	157.7 (180.2)
t*+1	727.5 (358.6)**	210.1 (45.0)***	132.4 (145.3)	1.8 (1.2)	$(208.8)^{**}$
t*+2	1330.9 (517.9)**	237 (59.4)***	336 (212.0)	1.3 (1.8)	639.9 (506.4)
t*+3	$(715,5)^{(110)}$	193.5 (77 4)**	(343.7)	(1.0) (2.0)	145.5 (642.1)
t*+4	(110.0) 1981.2 $(971.1)^{**}$	(11.4) 253.9 $(106.4)^{**}$	(535.0) 570.2 (525.7)	(2.0) 3.9 (3.0)	(642.1) -309.4 (684.2)
N. observation max N. observation min	694 539	694 539	694 539	446 310	446 310
mean of variables at t*-1 $$	1506	281	814	10.9	2354

All the specifications include the following controls: firm fixed effects, year dummies, age at t\*-1, age at t\*-1\*post, industry, industry\*post, area, area\*post. \* p < 0.1; \*\* p < 0.05; \*\*\* p < 0.01; standard errors are clustered at firm level. The means of variables at t\*-1 refer to the sample of venture-backed firms.

Post-treatment periods	$\begin{array}{c} \text{EBITDA/Assets} \\ \% \end{array}$	ROE %	Rating index	Patents dummy	Patents numbers	Survival rate
average post-treatment	-23.8 (6.4)***	23.2 (69.3)	0.6 (0.2)***	$0.04 \\ (0.04)$	$0.25 \\ (0.14)^*$	-0.008 (0.75)
$t^*$ (year of financing)	-20.5 (6.0)***	108.2 (90.5)	0.5 (0.2)**	-0.04 (0.02)*	-0.06 (0.07)	
t*+1	$(7.9)^{(310)}$	56.3 (89.4)	0.6 $(0.2)^{***}$	-0.01 (0.04)	0.07 (0.12)	
$t^*+2$	-22.9 (11.6)**	-201.8	0.6 (0.3)**	0.09 (0.06)	0.35	
t*+3	-16.3 (6.2)***	241.8 (127.6)*	0.8 (0.5)	0.19 $(0.07)^{**}$	$(0.31)^{***}$	
t*+4	(8.1)	-6.8 (39.5)	(0.10) (0.5)	$(0.09)^{**}$	$(0.43)^{***}$	
N. observation max N. observation min	692 538	$\begin{array}{c} 640 \\ 490 \end{array}$	649 492	$694 \\ 539$	$694 \\ 539$	293
mean of variables at t*-1	-11.5	-59.1	6.5	0.12	0.3	1

Table 5: Effects of venture capitalists on firms' profitability, innovation and survivorship Diff-in-diff estimations (coefficient  $beta_3$  is reported) - different post-treatment periods

All the specifications include the following controls: firm fixed effects, year dummies, age at t\*-1, age at t\*-1\*post, industry, industry\*post, area, area\*post. \* p < 0.1; \*\* p < 0.05; \*\*\* p < 0.01; standard errors are clustered at firm level. The means of variables at t\*-1 refer to the sample of venture-backed firms. Patents number is the cumulative number of patent applications in the period considered. Survival rate is the rate of survival after 3 years since financing/rejection, considering only firm financed until the year 2012; the mean at t\*-1 is 1 per cent as all firms are alive at that time.

#### Table 6: Effects of venture capitalists on firms' financial structure

Diff-in-diff estimations (coefficient  $beta_3$  is reported) - different post-treatment periods

Post-treatment periods	Leverage %	Fin. Debts (*1000 euro)	Equity (*1000 euro)	$\frac{\rm Bank/Fin.Debts}{\%}$	$\begin{array}{c} \text{Bank short/Bank} \\ \% \end{array}$	Interest rate $\%$
average post-treatment	-64.5 (38.9)*	165.1 (167.8)	452.1 (200.6)**	-5.6 (5.1)	$10.6 (5.4)^*$	7.1 (2.8)**
$t^*$ (year of financing)	-41.2 (40.9)	53.7 (156)	418.4 (115.2)***	0.14 (4.5)	5.7 (5.4)	5.9 (2.8)**
t*+1	-88.1 $(41.9)^{**}$	179.4 (170.6)	439.7 (196.5)**	-5.4 (6.0)	7.4 (7.0)	7.0 (3.6)*
$t^*+2$	-113.3 (53.9)**	218.6 (274.8)	766.7 (244.0)***	-9.7 (9.2)*	17.3 (9.2)	7.5 (4.8)
t*+3	-53.7 (34.4)	572.7 (341.9)*	801.4 (350.2)**	-8.9 (9.0)	23.5 (10.2)**	4.1 (3.4)
$t^{*}+4$	-61.3 (42.7)	127.3 (322.3)	1197.4 (628.4)*	-15.2 (11.9)	30.7 (11.8)***	2.3 (2.6)
N. observation max N. observation min	$\begin{array}{c} 618 \\ 468 \end{array}$	629 483	694 539	527 413	425 334	526 413
mean of variables at t*-1	96.6	544	393	57.5	75.2	4.5

All the specifications include the following controls: firm fixed effects, year dummies, age at t\*-1, age at t\*-1\*post, industry, industry\*post, area, area\*post. \* p < 0.1; \*\* p < 0.05; \*\*\* p < 0.01; standard errors are clustered at firm level. The means of variables at t\*-1 refer to the sample of venture-backed firms. Leverage is financial debts/(financial debts + equity); bank stands for bank loans; bank short are bank loans with maturity shorter than 1 year. Interest rate is financial costs on financial debts.

Table 7: Effects of venture capitalists on firms' size and activity indicators

Diff-in-diff estimations (coefficient  $beta_3$  is reported) - different post-treatment periods Specifications with a control sample of firms that increase their capital

Post-treatment periods	Assets (*1000 euro)	Labor costs (*1000 euro)	Sales (*1000 euro)	Employees number	Monthly wage euro
average post-treatment	965.7 $(418.3)^{**}$	147.2 (46.1)***	206.7 (184.2)	$1.5 (1.2)^*$	$192.3 \\ (242.1)$
$t^*$ (year of financing)	755.6 (391.7)*	74.2 (23.2)***	-31.9 (145.8)	3.3 (1.5)**	206.4 (200.9)
t*+1	931.2 (396.8)**	205.8 (44.4)***	192.5 (164.9)	2.0 (1.3)	392.6 (221.8)*
$t^*+2$	1617.6 (503.7)***	223.8 (63.7)***	347.8 (223.5)	-0.7 (2.7)	619.4 (613.6)
t*+3	1930.6 (771.6)**	170.2 (92.2)*	381.0 (402.5)	-1.4 (2.1)	-136.9 (620.9)
t*+4	2453.5 (1042.4)**	265.5 (122.2)**	$895.8 \\ (574.6)$	4.2 (3.5)	-470.2 (661.5)
N. observation max N. observation min	$522 \\ 395$	$522 \\ 395$	$\frac{522}{395}$	$330 \\ 227$	330 227
mean of variables at t*-1	1506	281	814	10.9	2354

All the specifications include the following controls: firm fixed effects, year dummies, age at t\*-1, age at t\*-1\*post, industry, industry\*post, area, area\*post. \* p < 0.1; \*\* p < 0.05; \*\*\* p < 0.01; standard errors are clustered at firm level. The means of variables at t\*-1 refer to the sample of venture-backed firms

Diff-in-diff estimations (coefficient  $beta_3$  is reported) - different post-treatment periods Specifications with a control sample of firms that increase their capital

Table 8: Effects of venture capitalists on firms' profitability, innovation and survivorship

Post-treatment periods	$\begin{array}{c} \text{EBITDA/Assets} \\ \% \end{array}$	ROE %	Rating index	Patents dummy	Patents numbers	Survival rate
average post-treatment	$^{-24.4}_{(6.5)***}$	11.3     (78.8)	0.7 (0.2)***	$\begin{array}{c} 0.05 \ (0.02) \end{array}$	$0.26 \\ (0.16)^*$	-0.08 (0.079)
$t^*$ (year of financing)	-21.0 (6.2)***	79.3 (111.7)	0.5 (0.2)**	-0.03 (0.03)	-0.05 (0.07)	
$t^*+1$	-25.2 (8.1)***	143.0 (160.8)	0.6 (0.3)**	0.00 (0.04)	0.09 (0.13)	
$t^*+2$	-23.4 (10.7)**	-184.9 (305.9)	0.7 (0.3)**	0.10 (0.06)	0.3 (0.18)*	
t*+3	-17.5 (5.9)***	279.9 (145.5)*	0.8 (0.5)	0.20 (0.08)**	1.1 (0.3)***	
t*+4	6.0 (9.4)	-8.9 (47.2)	0.0 (0.5)	0.21 (0.09)**	1.5 (0.5)***	
N. observation max N. observation min	$520\\394$	$473 \\ 349$	482 353	$522 \\ 395$	522 395	208
mean of variables at t*-1	-11.5	-59.1	6.5	0.12	0.3	1

All the specifications include the following controls: firm fixed effects, year dummies, age at t\*-1, age at t\*-1\*post, industry, industry\*post, area, area\*post. \* p < 0.1; \*\* p < 0.05; \*\*\* p < 0.01; standard errors are clustered at firm level. The means of variables at t\*-1 refer to the sample of venture-backed firms. Patents number is the cumulative number of patent applications in the period considered. Survival rate is the rate of survival after 3 years since financing/rejection, considering only firm financed until the year 2012; the mean at t\*-1 is 1 per cent as all firms are alive at that time.

Table 9: Effects of venture capitalists on firms' financial structure

Diff-in-diff estimations (coefficient  $beta_3$  is reported) - different post-treatment periods Specifications with a control sample of firms that increase their capital

Post-treatment periods	Leverage $\%$	Fin. Debts (*1000 euro)	Equity (*1000 euro)	$\frac{\rm Bank/Fin.Debts}{\%}$	Bank short/Bank $\%$	Interest rate $\%$
average post-treatment	-63.6 (40.4)	223.0 (187.1)	447.9 (215.6)**	-7.9 (5.4)	9.6 (5.9)	7.3 (2.9)**
$t^*$ (year of financing)	-36.5 (41.6)	118.7 (177.9)	415.2 (126.8)***	-1.0 (5.2)	7.2 (5.9)	6.6 (3.0)**
$t^*+1$	-83.8 (41.9)**	219.1 (194.5)	450.6 (212.5)**	-9.9 (6.0)	5.5 (7.7)	5.7 (3.4)*
$t^*+2$	-123.5 (55.8)**	297.8 (270.3)	786.1 (255.7)***	-12.8 (9.5)	14.5 (11.0)	8.1 (5.0)
t*+3	-63.3 (41.5)	584.8 (374.9)	825.1 (361.6)**	-8.8 (9.2)	18.6 (12.3)	5.4 (3.8)
t*+4	-64.7 (51.4)	9.4 $(377.3)$	1317.2 (641.1)**	-11.7 (12.2)	$(13.4)^*$	3.3 (2.3)
N. observation max N. observation min	468 347	478 361	522 395	401 313	321 248	400 313
mean of variables at t*-1	96.6	543	393	57.5	75.2	4.5

All the specifications include the following controls: firm fixed effects, year dummies, age at t\*-1, age at t\*-1\*post, industry, industry\*post, area, area\*post. \* p < 0.1; \*\* p < 0.05; \*\*\* p < 0.01; standard errors are clustered at firm level. The means of variables at t\*-1 refer to the sample of venture-backed firms. Leverage is financial debts/(financial debts + equity); bank stands for bank loans; bank short are bank loans with maturity shorter than 1 year. Interest rate is financial costs on financial debts.

Table 10: Effects of venture capitalists on firms' size and activity indicators

Diff-in-diff estimations (coefficient  $beta_3$  is reported) - different post-treatment periods Specifications that split between independent and captive venture capitalists

Post-treatment periods	Assets (*1000 euro)	Labor costs (*1000 euro)	Sales (*1000 euro)	Employees number	Monthly wage euro
average post-treatment independent	698.4**	210.5***	151.4	5.4***	182.1
average post-treatment captive	897.1	81.6	89.8	-1.1	196.5
t <sup>*</sup> (year of financing) independent	$378.4^{*}$	93.3***	-25.0	5.7***	253.2
t <sup>*</sup> (year of financing) captive	933.0	85.1**	-36.5	2.8	69.0
$t^{*}+1$ independent	707.0**	$263.1^{***}$	115.9	$3.5^{***}$	634.2**
$t^*+1$ captive	756.2	$135.6^{**}$	155.6	-0.0	198.7
$t^*+2$ independent	$1213.8^{***}$	$320.1^{***}$	304.7	$4.2^{*}$	219.1
$t^*+2$ captive	1500.8	116.1	381.4	-2.6	1230.2
$t^{*}+3$ independent	$1699.3^{***}$	$334.5^{***}$	512.0	$6.2^{***}$	646.5
$t^*+3$ captive	1750.2	-9.0	102.0	-7.2***	-430.8
$t^*+4$ independent	$1779.9^{*}$	381.3***	687.0	$9.7^{***}$	232.3
$t^{*}+4$ captive	2300.4	51.9	384.9	-10.6***	-1694.9**
N. observation max	694	694	694	446	446
N. observation min	539	539	539	310	310
mean of variables at t*-1	1506	281	814	10.9	2354

All the specifications include the following controls: firm fixed effects, year dummies, age at t\*-1, age at t\*-1\*post, industry, industry\*post, area, area\*post. \* p < 0.1; \*\* p < 0.05; \*\*\* p < 0.01; standard errors are clustered at firm level and are not reported to preserve space. The means of variables at t\*-1 refer to the sample of venture-backed firms

Table 11: Effects of venture capitalists on firms' profitability, innovation and survivorship

Diff-in-diff estimations (coefficient  $beta_3$  is reported) - different post-treatment periods Specifications with a control sample of firms that increase their capital

Post-treatment periods	$\begin{array}{c} \text{EBITDA/Assets} \\ \% \end{array}$	ROE %	Rating index	Patents dummy	Patents numbers	Survival rate
average post-treatment independent average post-treatment captive	-28.4*** -17.4**	$30.3 \\ 11.2$	$0.6^{**}$ $0.7^{***}$	$0.02 \\ 0.06$	$0.33^{*}$ 0.14	0.03 -0.06
t* (year of financing) independent t* (year of financing) captive t*+1 independent t*+1 captive t*+2 independent t*+2 captive t*+3 independent t*+3 captive t*+4 independent t*+4 captive	$\begin{array}{c} -21.6^{***} \\ -18.9^{*} \\ -26.9^{**} \\ -20.2^{*} \\ -40.6^{**} \\ 3.0 \\ -22.3^{**} \\ -7.3 \\ 3.7 \\ 4.0 \end{array}$	$\begin{array}{c} 109.1 \\ 106.5 \\ 135.0 \\ -81.6 \\ -450.5 \\ 231.9 \\ -217.1 \\ -277.7^* \\ 39.0 \\ -95.4 \end{array}$	$\begin{array}{c} 0.5^{*} \\ 0.5^{*} \\ 0.6^{*} \\ 0.7^{*} \\ 0.6 \\ 0.5 \\ 1.2^{**} \\ -0.4 \\ 0.9 \end{array}$	$\begin{array}{c} -0.03\\ -0.05^{*}\\ -0.03\\ 0.02\\ 0.05\\ 0.14\\ 0.17^{*}\\ 0.21^{*}\\ 0.20^{*}\\ 0.22\end{array}$	$\begin{array}{c} -0.01 \\ -0.12^{**} \\ 0.07 \\ 0.07 \\ 0.42 \\ 0.26 \\ 1.5^{***} \\ 0.5 \\ 2.0^{***} \\ 0.9 \end{array}$	
N. observation max N. observation min	692 538	640 490	649 492	$694 \\ 539$	$694 \\ 539$	293
mean of variables at t*-1	-11.5	-59.1	6.5	0.12	0.3	1

All the specifications include the following controls: firm fixed effects, year dummies, age at t\*-1, age at t\*-1\*post, industry, industry\*post, area, area\*post. \* p < 0.1; \*\* p < 0.05; \*\*\* p < 0.01; standard errors are clustered at firm level and are not reported to preserve space. The means of variables at t\*-1 refer to the sample of venture-backed firms. Patents number is the cumulative number of patent applications in the period considered. Survival rate is the rate of survival after 3 years since financing/rejection, considering only firm financed until the year 2012; the mean at t\*-1 is 1 per cent as all firms are alive at that time.

#### Table 12: Effects of venture capitalists on firms' financial structure

Diff-in-diff estimations (coefficient  $beta_3$  is reported) - different post-treatment periods Specifications that split between independent and captive venture capitalists

Post-treatment periods	Leverage %	Fin. Debts (*1000 euro)	Equity (*1000 euro)	$\frac{\rm Bank/Fin.Debts}{\%}$	$\begin{array}{c} \text{Bank short/Bank} \\ \% \end{array}$	Interest rate $\%$
average post-treatment independent	-61.8**	-63.5	582.5**	-9.8	5.6	9.7**
average post-treatment captive	-68.0	462.7	266.3	-0.5	$16.1^{**}$	3.8
t* (year of financing) independent	-43.8	-162.5	493.3***	-5.0	3.0	7.8**
t <sup>*</sup> (year of financing) captive	-37.7	335.2	$311.7^{**}$	6.7	8.6	3.6
$t^{*}+1$ independent	-73.9**	-22.8	$667.4^{**}$	0.0	2.6	7.9
$t^{*}+1$ captive	-110.3	480.2	120.1	-1.0	$13.7^{*}$	5.7
$t^{*}+2$ independent	-67.1**	-154.7	$977.1^{***}$	-12.4	13.3	1.6
$t^{*}+2$ captive	-182	763.6	$460.9^{*}$	5.9	$24.2^{*}$	15.7
$t^{*}+3$ independent	-76.9*	149.2	$1068.7^{*}$	-16.6	20.9	4.6
$t^{*}+3$ captive	-19.1	$1189.6^{*}$	417.6	1.3	27.2**	3.3
$t^{*}+4$ independent	-84.9*	-108.2	$1413.7^{**}$	-17.6	27.5**	3.2
$t^*+4$ captive	-28.2	463.5	854.4	-11.7	37.3**	1.0
N. observation max	618	629	694	527	425	526
N. observation min	468	483	539	413	334	413
mean of variables at t*-1	96.6	543	393	57.5	75.2	4.5

All the specifications include the following controls: firm fixed effects, year dummies, age at t\*-1, age at t\*-1\*post, industry, industry\*post, area, area\*post. \* p < 0.1; \*\* p < 0.05; \*\*\* p < 0.01; standard errors are clustered at firm level and are not reported to preserve space. The means of variables at t\*-1 refer to the sample of venture-backed firms. Leverage is financial debts/(financial debts + equity); bank stands for bank loans; bank short are bank loans with maturity shorter than 1 year. Interest rate is financial costs on financial debts.

#### RECENTLY PUBLISHED "TEMI" (\*)

- N. 1110 Services trade and credit frictions: evidence from matched bank-firm data, by Francesco Bripi, David Loschiavo and Davide Revelli (April 2017).
- N. 1111 Public guarantees on loans to SMEs: an RDD evaluation, by Guido de Blasio, Stefania De Mitri, Alessio D'Ignazio, Paolo Finaldi Russo and Lavina Stoppani (April 2017).
- N. 1112 Local labour market heterogeneity in Italy: estimates and simulations using responses to labour demand shocks, by Emanuele Ciani, Francesco David and Guido de Blasio (April 2017).
- N. 1113 Liquidity transformation and financial stability: evidence from the cash management of open-end Italian mutual funds, by Nicola Branzoli and Giovanni Guazzarotti (April 2017).
- N. 1114 Assessing the risks of asset overvaluation: models and challenges, by Sara Cecchetti and Marco Taboga (April 2017).
- N.1115 Social ties and the demand for financial services, by Eleonora Patacchini and Edoardo Rainone (June 2017).
- N.1116 Measurement errors in consumption surveys and the estimation of poverty and inequality indices, by Giovanni D'Alessio (June 2017).
- N.1117 No free lunch, Buddy: past housing transfers and informal care later in life, by Emanuele Ciani and Claudio Deiana (June 2017).
- N. 1118 *The interbank network across the global financial crisis: evidence from Italy*, by Massimiliano Affinito and Alberto Franco Pozzolo (June 2017).
- N. 1119 *The collateral channel of unconventional monetary policy*, by Giuseppe Ferrero, Michele Loberto and Marcello Miccoli (June 2017).
- N.1120 Medium and long term implications of financial integration without financial development, by Flavia Corneli (June 2017).
- N. 1121 *The financial stability dark side of monetary policy*, by Piergiorgio Alessandri, Antonio Maria Conti and Fabrizio Venditti (June 2017).
- N.1122 Large time-varying parameter VARs: a non-parametric approach, by George Kapetanios, Massimiliano Marcellino and Fabrizio Venditti (June 2017).
- N. 1123 *Multiple lending, credit lines, and financial contagion*, by Giuseppe Cappelletti and Paolo Emilio Mistrulli (June 2017).
- N. 1124 *Law enforcement and political participation: Italy, 1861-65*, by Antonio Accetturo, Matteo Bugamelli and Andrea Lamorgese (July 2017).
- N. 1125 The consequences of public employment: evidence from Italian municipalities, by Marta Auricchio, Emanuele Ciani, Alberto Dalmazzo and Guido de Blasio (July 2017).
- N. 1126 *The cyclicality of the income elasticity of trade*, by Alessandro Borin, Virginia Di Nino, Michele Mancini and Massimo Sbracia (July 2017).
- N. 1127 Human capital and urban growth in Italy, 1981-2001, by Francesco Giffoni, Matteo Gomellini and Dario Pellegrino (July 2017).
- N. 1128 *The double bind of asymmetric information in over-the-counter markets*, by Taneli Mäkinen and Francesco Palazzo (July 2017).
- N. 1129 *The effects of central bank's verbal guidance: evidence from the ECB*, by Maddalena Galardo and Cinzia Guerrieri (July 2017).
- N. 1130 The Bank of Italy econometric model: an update of the main equations and model elasticities, by Guido Bulligan, Fabio Busetti, Michele Caivano, Pietro Cova, Davide Fantino, Alberto Locarno, Lisa Rodano (July 2017).

<sup>(\*)</sup> Requests for copies should be sent to:

Banca d'Italia – Servizio Studi di struttura economica e finanziaria – Divisione Biblioteca e Archivio storico – Via Nazionale, 91 – 00184 Rome – (fax 0039 06 47922059). They are available on the Internet www.bancaditalia.it.

- ALBERTAZZI U., G. ERAMO, L. GAMBACORTA and C. SALLEO, Asymmetric information in securitization: an empirical assessment, Journal of Monetary Economics, v. 71, pp. 33-49, TD No. 796 (February 2011).
- ALESSANDRI P. and B. NELSON, *Simple banking: profitability and the yield curve,* Journal of Money, Credit and Banking, v. 47, 1, pp. 143-175, **TD No. 945 (January 2014).**
- ANTONIETTI R., R. BRONZINI and G. CAINELLI, *Inward greenfield FDI and innovation*, Economia e Politica Industriale, v. 42, 1, pp. 93-116, **TD No. 1006** (March 2015).
- BARONE G. and G. NARCISO, Organized crime and business subsidies: Where does the money go?, Journal of Urban Economics, v. 86, pp. 98-110, **TD No. 916** (June 2013).
- BRONZINI R., The effects of extensive and intensive margins of FDI on domestic employment: microeconomic evidence from Italy, B.E. Journal of Economic Analysis & Policy, v. 15, 4, pp. 2079-2109, TD No. 769 (July 2010).
- BUGAMELLI M., S. FABIANI and E. SETTE, The age of the dragon: the effect of imports from China on firmlevel prices, Journal of Money, Credit and Banking, v. 47, 6, pp. 1091-1118, TD No. 737 (January 2010).
- BULLIGAN G., M. MARCELLINO and F. VENDITTI, *Forecasting economic activity with targeted predictors,* International Journal of Forecasting, v. 31, 1, pp. 188-206, **TD No. 847 (February 2012).**
- BUSETTI F., On detecting end-of-sample instabilities, in S.J. Koopman, N. Shepard (eds.), Unobserved Components and Time Series Econometrics, Oxford, Oxford University Press, TD No. 881 (September 2012).
- CESARONI T., *Procyclicality of credit rating systems: how to manage it*, Journal of Economics and Business, v. 82. pp. 62-83, **TD No. 1034** (October 2015).
- CIARLONE A., *House price cycles in emerging economies*, Studies in Economics and Finance, v. 32, 1, **TD No. 863 (May 2012).**
- CUCINIELLO V. and F. M. SIGNORETTI, *Large banks,loan rate markup and monetary policy*, International Journal of Central Banking, v. 11, 3, pp. 141-177, **TD No. 987** (November 2014).
- DE BLASIO G., D. FANTINO and G. PELLEGRINI, *Evaluating the impact of innovation incentives: evidence from an unexpected shortage of funds*, Industrial and Corporate Change, v. 24, 6, pp. 1285-1314, **TD No. 792 (February 2011).**
- DEPALO D., R. GIORDANO and E. PAPAPETROU, *Public-private wage differentials in euro area countries:* evidence from quantile decomposition analysis, Empirical Economics, v. 49, 3, pp. 985-1115, **TD No. 907 (April 2013).**
- DI CESARE A., A. P. STORK and C. DE VRIES, *Risk measures for autocorrelated hedge fund returns*, Journal of Financial Econometrics, v. 13, 4, pp. 868-895, **TD No. 831 (October 2011).**
- FANTINO D., A. MORI and D. SCALISE, Collaboration between firms and universities in Italy: the role of a firm's proximity to top-rated departments, Rivista Italiana degli economisti, v. 1, 2, pp. 219-251, TD No. 884 (October 2012).
- FRATZSCHER M., D. RIMEC, L. SARNOB and G. ZINNA, *The scapegoat theory of exchange rates: the first tests*, Journal of Monetary Economics, v. 70, 1, pp. 1-21, **TD No. 991 (November 2014).**
- NOTARPIETRO A. and S. SIVIERO, *Optimal monetary policy rules and house prices: the role of financial frictions,* Journal of Money, Credit and Banking, v. 47, S1, pp. 383-410, **TD No. 993 (November 2014).**
- RIGGI M. and F. VENDITTI, *The time varying effect of oil price shocks on euro-area exports*, Journal of Economic Dynamics and Control, v. 59, pp. 75-94, **TD No. 1035 (October 2015).**
- TANELI M. and B. OHL, *Information acquisition and learning from prices over the business cycle*, Journal of Economic Theory, 158 B, pp. 585–633, **TD No. 946 (January 2014).**

- ALBANESE G., G. DE BLASIO and P. SESTITO, *My parents taught me. evidence on the family transmission of values,* Journal of Population Economics, v. 29, 2, pp. 571-592, **TD No. 955 (March 2014).**
- ANDINI M. and G. DE BLASIO, *Local development that money cannot buy: Italy's Contratti di Programma,* Journal of Economic Geography, v. 16, 2, pp. 365-393, **TD No. 915 (June 2013).**
- BARONE G. and S. MOCETTI, *Inequality and trust: new evidence from panel data*, Economic Inquiry, v. 54, pp. 794-809, **TD No. 973 (October 2014).**
- BELTRATTI A., B. BORTOLOTTI and M. CACCAVAIO, Stock market efficiency in China: evidence from the split-share reform, Quarterly Review of Economics and Finance, v. 60, pp. 125-137, TD No. 969 (October 2014).
- BOLATTO S. and M. SBRACIA, *Deconstructing the gains from trade: selection of industries vs reallocation of workers*, Review of International Economics, v. 24, 2, pp. 344-363, **TD No. 1037 (November 2015).**
- BOLTON P., X. FREIXAS, L. GAMBACORTA and P. E. MISTRULLI, *Relationship and transaction lending in a crisis*, Review of Financial Studies, v. 29, 10, pp. 2643-2676, **TD No. 917 (July 2013).**
- BONACCORSI DI PATTI E. and E. SETTE, Did the securitization market freeze affect bank lending during the financial crisis? Evidence from a credit register, Journal of Financial Intermediation, v. 25, 1, pp. 54-76, TD No. 848 (February 2012).
- BORIN A. and M. MANCINI, Foreign direct investment and firm performance: an empirical analysis of *Italian firms*, Review of World Economics, v. 152, 4, pp. 705-732, **TD No. 1011 (June 2015).**
- BRAGOLI D., M. RIGON and F. ZANETTI, *Optimal inflation weights in the euro area*, International Journal of Central Banking, v. 12, 2, pp. 357-383, **TD No. 1045** (January 2016).
- BRANDOLINI A. and E. VIVIANO, Behind and beyond the (headcount) employment rate, Journal of the Royal Statistical Society: Series A, v. 179, 3, pp. 657-681, TD No. 965 (July 2015).
- BRIPI F., *The role of regulation on entry: evidence from the Italian provinces*, World Bank Economic Review, v. 30, 2, pp. 383-411, **TD No. 932 (September 2013).**
- BRONZINI R. and P. PISELLI, *The impact of R&D subsidies on firm innovation*, Research Policy, v. 45, 2, pp. 442-457, **TD No. 960 (April 2014).**
- BURLON L. and M. VILALTA-BUFI, A new look at technical progress and early retirement, IZA Journal of Labor Policy, v. 5, **TD No. 963 (June 2014).**
- BUSETTI F. and M. CAIVANO, The trend-cycle decomposition of output and the Phillips Curve: bayesian estimates for Italy and the Euro Area, Empirical Economics, V. 50, 4, pp. 1565-1587, TD No. 941 (November 2013).
- CAIVANO M. and A. HARVEY, *Time-series models with an EGB2 conditional distribution*, Journal of Time Series Analysis, v. 35, 6, pp. 558-571, **TD No. 947** (January 2014).
- CALZA A. and A. ZAGHINI, *Shoe-leather costs in the euro area and the foreign demand for euro banknotes,* International Journal of Central Banking, v. 12, 1, pp. 231-246, **TD No. 1039 (December 2015).**
- CIANI E., *Retirement, Pension eligibility and home production*, Labour Economics, v. 38, pp. 106-120, **TD** No. 1056 (March 2016).
- CIARLONE A. and V. MICELI, Escaping financial crises? Macro evidence from sovereign wealth funds' investment behaviour, Emerging Markets Review, v. 27, 2, pp. 169-196, TD No. 972 (October 2014).
- CORNELI F. and E. TARANTINO, *Sovereign debt and reserves with liquidity and productivity crises*, Journal of International Money and Finance, v. 65, pp. 166-194, **TD No. 1012** (June 2015).
- D'AURIZIO L. and D. DEPALO, An evaluation of the policies on repayment of government's trade debt in *Italy*, Italian Economic Journal, v. 2, 2, pp. 167-196, **TD No. 1061 (April 2016).**
- DE BLASIO G., G. MAGIO and C. MENON, Down and out in Italian towns: measuring the impact of economic downturns on crime, Economics Letters, 146, pp. 99-102, TD No. 925 (July 2013).
- DOTTORI D. and M. MANNA, *Strategy and tactics in public debt management*, Journal of Policy Modeling, v. 38, 1, pp. 1-25, **TD No. 1005 (March 2015).**
- ESPOSITO L., A. NOBILI and T. ROPELE, *The management of interest rate risk during the crisis: evidence from Italian banks*, Journal of Banking & Finance, v. 59, pp. 486-504, **TD No. 933 (September 2013).**
- MARCELLINO M., M. PORQUEDDU and F. VENDITTI, *Short-Term GDP forecasting with a mixed frequency dynamic factor model with stochastic volatility*, Journal of Business & Economic Statistics, v. 34, 1, pp. 118-127, **TD No. 896 (January 2013).**

- RODANO G., N. SERRANO-VELARDE and E. TARANTINO, *Bankruptcy law and bank financing*, Journal of Financial Economics, v. 120, 2, pp. 363-382, **TD No. 1013 (June 2015).**
- ZINNA G., *Price pressures on UK real rates: an empirical investigation*, Review of Finance, v. 20, 4, pp. 1587-1630, **TD No. 968 (July 2014).**

2017

- ADAMOPOULOU A. and G.M. TANZI, Academic dropout and the great recession, Journal of Human Capital, V. 11, 1, pp. 35–71, **TD No. 970 (October 2014).**
- ALBERTAZZI U., M. BOTTERO and G. SENE, Information externalities in the credit market and the spell of credit rationing, Journal of Financial Intermediation, v. 30, pp. 61–70, TD No. 980 (November 2014).
- ALESSANDRI P. and H. MUMTAZ, *Financial indicators and density forecasts for US output and inflation*, Review of Economic Dynamics, v. 24, pp. 66-78, **TD No. 977 (November 2014).**
- BRUCHE M. and A. SEGURA, *Debt maturity and the liquidity of secondary debt markets*, Journal of Financial Economics, v. 124, 3, pp. 599-613, **TD No. 1049 (January 2016).**
- DE BLASIO G. and S. POY, *The impact of local minimum wages on employment: evidence from Italy in the* 1950s, Journal of Regional Science, v. 57, 1, pp. 48-74, **TD No. 953 (March 2014).**
- LOBERTO M. and C. PERRICONE, *Does trend inflation make a difference?*, Economic Modelling, v. 61, pp. 351–375, **TD No. 1033 (October 2015).**
- MOCETTI S., M. PAGNINI and E. SETTE, *Information technology and banking organization*, Journal of Journal of Financial Services Research, v. 51, pp. 313-338, **TD No. 752** (March 2010).
- MOCETTI S. and E. VIVIANO, *Looking behind mortgage delinquencies*, Journal of Banking & Finance, v. 75, pp. 53-63, **TD No. 999 (January 2015).**
- PALAZZO F., Search costs and the severity of adverse selection, Research in Economics, v. 71, 1, pp. 171-197, **TD No. 1073 (July 2016).**
- PATACCHINI E., E. RAINONE and Y. ZENOU, *Heterogeneous peer effects in education*, Journal of Economic Behavior & Organization, v. 134, pp. 190–227, **TD No. 1048** (January 2016).

#### FORTHCOMING

- ADAMOPOULOU A. and E. KAYA, Young Adults living with their parents and the influence of peers, Oxford Bulletin of Economics and Statistics, **TD No. 1038** (November 2015).
- BOFONDI M., L. CARPINELLI and E. SETTE, *Credit supply during a sovereign debt crisis*, Journal of the European Economic Association, **TD No. 909** (April 2013).
- BRONZINI R. and A. D'IGNAZIO, *Bank internationalisation and firm exports: evidence from matched firmbank data*, Review of International Economics, **TD No. 1055 (March 2016).**
- BURLON L., Public expenditure distribution, voting, and growth, Journal of Public Economic Theory, TD No. 961 (April 2014).
- BUSETTI F., *Quantile aggregation of density forecasts*, Oxford Bulletin of Economics and Statistics, **TD No. 979 (November 2014).**
- CESARONI T. and R. DE SANTIS, *Current account "core-periphery dualism" in the EMU*, World Economy, **TD No. 996 (December 2014).**
- CESARONI T. and S. IEZZI, *The predictive content of business survey indicators: evidence from SIGE,* Journal of Business Cycle Research, **TD No. 1031 (October 2015).**
- CONTI P., D. MARELLA and A. NERI, *Statistical matching and uncertainty analysis in combining household income and expenditure data*, Statistical Methods & Applications, **TD No. 1018 (July 2015).**
- D'AMURI F., Monitoring and disincentives in containing paid sick leave, Labour Economics, TD No. 787 (January 2011).
- D'AMURI F. and J. MARCUCCI, *The predictive power of google searches in forecasting unemployment*, International Journal of Forecasting, **TD No. 891 (November 2012).**

- FEDERICO S. and E. TOSTI, *Exporters and importers of services: firm-level evidence on Italy*, The World Economy, **TD No. 877 (September 2012).**
- GIACOMELLI S. and C. MENON, *Does weak contract enforcement affect firm size? Evidence from the neighbour's court,* Journal of Economic Geography, **TD No. 898 (January 2013).**
- MANCINI A.L., C. MONFARDINI and S. PASQUA, *Is a good example the best sermon? Children's imitation of parental reading*, Review of Economics of the Household, **D No. 958 (April 2014).**
- MEEKS R., B. NELSON and P. ALESSANDRI, *Shadow banks and macroeconomic instability*, Journal of Money, Credit and Banking, **TD No. 939** (November 2013).
- MICUCCI G. and P. ROSSI, *Debt restructuring and the role of banks' organizational structure and lending technologies*, Journal of Financial Services Research, **TD No. 763 (June 2010).**
- NATOLI F. and L. SIGALOTTI, *Tail co-movement in inflation expectations as an indicator of anchoring,* International Journal of Central Banking, **TD No. 1025 (July 2015).**
- RIGGI M., Capital destruction, jobless recoveries, and the discipline device role of unemployment, Macroeconomic Dynamics, **TD No. 871 July 2012**).
- SEGURA A., Why did sponsor banks rescue their SIVs?, Review of Finance, TD No. 1100 (February 2017).
- SEGURA A. and J. SUAREZ, *How excessive is banks' maturity transformation?*, Review of Financial Studies, **TD No. 1065 (April 2016).**